Social Science Research xxx (2011) xxx-xxx

ELSEVIER

Contents lists available at ScienceDirect

# Social Science Research

journal homepage: www.elsevier.com/locate/ssresearch

# The Earned Income Tax Credit and abortion

## Chris M. Herbst

School of Public Affairs, Arizona State University, Mail Code 3720, 411 N. Central Ave., Ste. 450, Phoenix, AZ 85004-0687, United States

## ARTICLE INFO

Article history: Received 25 August 2010 Revised 12 April 2011 Accepted 23 May 2011 Available online xxxx

*Keywords:* Abortion Fertility Earned Income Tax Credit Panel data methods

## ABSTRACT

Using a panel of states between 1975 and 2005, I examine the relationship between the Earned Income Tax Credit (EITC) and abortion. Results suggest that increases in the EITC are associated with reductions in the overall abortion rate. Specifically, a \$1000 increase in the maximum credit is associated with a 7.6% decrease in the overall abortion rate. This effect implies a reduction of approximately 1.6 abortions per 1000 women of childbearing age. I also provide evidence that the reduction in abortions is attributable to a drop in pregnancies, not an increase in the number of births. In other words, women appear to respond to increases in EITC generosity by altering sexual behavior, rather than shifting abortion–birth decisions after a pregnancy occurs.

© 2011 Elsevier Inc. All rights reserved.

## 1. Introduction

After peaking in 1980–7 years after the US Supreme Court decision in *Roe v. Wade*—the abortion rate declined substantially. By 2005, there were about 19 abortions per 1000 women of childbearing age, a 34% decrease from the early-1980s. Indeed, the abortion rate is currently at its lowest level since the Court's decision in 1973 (Finer and Henshaw, 2006). A sizable body of research has advanced a number of explanations for this decline, including changes in demographic trends and economic conditions and the enactment of public policies aimed at restricting women's access to abortion (Bitler and Zavodny, 2001; Blank et al., 1996; Kane and Staiger, 1996; Levine, 2003; Matthews et al., 1997).

Means-tested transfers can also influence women's fertility decisions because many of these programs target benefits at women with children. Considerable research focuses on the cash assistance program Aid to Families with Dependent Children/Temporary Assistance to Needy Families (AFDC/TANF), which not only conditions eligibility on having children but also provides families with greater benefits as the number of children increases (Moffitt, 1998).<sup>1</sup> Furthermore, the Earned Income Tax Credit (EITC), a wage supplement for low-income workers, potentially distorts fertility decisions for many of the same reasons. With annual expenditures and recipients that exceed the AFDC/TANF program, it is surprising that researchers have neglected a systematic examination of the relationship between the EITC and abortion.<sup>2</sup> The goal of this paper is to provide such an analysis.

There are several reasons to believe that the EITC may influence the decision of women to obtain an abortion. The first set of considerations deals with the program's design features. With the exception of a small credit for childless tax filers, the EITC is largely directed at families with children. This feature implies that the EITC lowers the relative cost of childbearing, leading to a possible reduction in the number of abortions and an increase in the number of births. Another design feature emphasizes variation in benefit generosity according to the number of children in the family. Although EITC benefits did not

E-mail address: chris.herbst@asu.edu

<sup>&</sup>lt;sup>1</sup> In response, a number of states implemented a family cap policy throughout the 1990s. This policy ends the incremental increase in AFDC/TANF benefits for each additional child born into a family. As of 2005, 22 states implemented a family cap.

<sup>&</sup>lt;sup>2</sup> As of 2005, federal expenditures through the EITC totaled \$38.8 billion, while spending on AFDC/TANF totaled \$24.8 billion. Approximately 22.8 million families claimed the credit in 2005. The comparable figure for the AFDC/TANF program is two million families (in 2003) (Urban-Brookings Tax Policy Center, 2008).

<sup>0049-089</sup>X/\$ - see front matter  $\odot$  2011 Elsevier Inc. All rights reserved. doi:10.1016/j.ssresearch.2011.05.003

2

#### C.M. Herbst/Social Science Research xxx (2011) xxx-xxx

initially vary with the number of children, a series of reforms enacted in 1990 and 1993 created a substantially greater maximum credit for families with two or more children compared to those with a single child. Therefore, prior to the early-1990s legislation, the EITC provided an unambiguous incentive to reduce abortions among childless women. Introduction of the differential benefit for families with two or more children, however, may have encouraged further reductions in abortions among one-child families.

A second set of considerations deals with the *timing* of the EITC's influence in the fertility decision-making process. In particular, the introduction of an EITC may alter abortion rates first through changes in sexual activity and contraceptive intensity, both of which ultimately affect pregnancy rates. As discussed in the next section, there are theoretical reasons to believe that the EITC can increase pregnancy rates within some groups of women while decreasing pregnancies within other groups. A key issue here is determining how those EITC-induced pregnancies are ultimately resolved by women. The EITC may also influence abortion rates by altering women's decision-making after a pregnancy occurs. However, theoretical models provide similarly ambiguous predictions for whether the policy will lead to an increase in the number of births or an increase in the number of abortions.

A final set of issues focuses on the complex employment incentives created by the EITC. If a family's earnings fall within the phase-in region of the benefit schedule, where the amount of the credit is rising with income, the policy creates strong employment incentives that may increase the opportunity costs associated with children. This implies that abortions may rise because of the EITC. For families located in the plateau region, in which the credit amount no longer increases with earnings, the ambiguous labor supply incentives suggest that only an income effect is relevant to decision-making. Assuming that children are viewed as normal goods, the income effect implies a reduction in the number of abortions. Finally, families located in the phase-out region experience employment disincentives because EITC benefits decline with earnings. In this region, the substitution effects (reduction in work) and income effects (increased income from the EITC) both imply reductions in the abortion rate. Together, these complicated employment incentives, along with the design features discussed above, suggest that the EITC has theoretically ambiguous effects on abortion decisions. Therefore, empirical analysis is necessary.

In this paper, I draw upon state-level panel data between 1975 and 2005 to examine the association between the EITC and abortion. A key advantage of the panel data approach is that it allows one to account for unobserved and difficult-to-measure determinants of state-level abortion rates that may otherwise bias the impact of the EITC. The analysis begins by using information from the Centers for Disease Control (CDC) to calculate the overall abortion rate, which is used in regressions on the combined federal and state EITC maximum credit. The key result emerging from this analysis is that increases in the EITC are associated with reductions in state-level abortion rates. Specifically, a \$1000 increase in the maximum credit is associated with a 7.6% decrease in the overall abortion rate. This effect implies a reduction of approximately 1.6 abortions per 1000 women of childbearing age. I then turn to an exploration of the mechanism through which the EITC influences abortion rates. In particular, I construct state-level birth and pregnancy rates to determine whether the reduction in abortions is due to increases in the number of births or reductions in the number of pregnancies. I find that the EITC is associated with declines in birth rates *and* pregnancy rates. This provides tentative evidence that the credit reduces abortions through changes in sexual and contraceptive behavior, rather than changes in the birth–abortion decision once a pregnancy occurs.

The remainder of the paper proceeds as follows. Section 2 provides an overview of the EITC and develops a simple theoretical framework to guide the empirical analysis. Section 3 summarizes what is currently known about the influence of public policies on women's fertility decisions. The data and estimation approach are introduced in Section 4. Section 5 discusses the results, and Section 6 concludes.

## 2. The Earned Income Tax Credit: structure and incentive effects

## 2.1. Historical development, current structure, and program awareness

The 1975 Tax Reduction Act established an earned income credit for tax filers with children. It was initially conceptualized to achieve three goals: act as a "work bonus" for the working poor, offset some of the growth in payroll taxes, and respond to the 1974 recession by stimulating demand. The credit provided a 10% wage subsidy on earnings up to \$4000 (for a maximum credit of \$400), which was then phased out at a rate of 10% until earnings reached \$8000. The EITC was structured as a refundable tax credit: if EITC benefits exceeded tax liabilities, families received a check for the difference from the Internal Revenue Service (IRS).

Major legislation on the EITC slowed for several years until the Tax Reform Act of 1986. This law raised the subsidy rate to 14% and increased the maximum credit to \$851. The EITC received its second major expansion through the Omnibus Budget Reconciliation Act of 1990. The defining feature of this legislation was the creation of separate benefit schedules for one- and multiple-child families. Those with one child could receive a 16.7% wage subsidy (for a maximum credit of \$1192), while families with two or more children could receive a 17.3% subsidy (for a maximum credit of \$1235). A third expansion occurred through the Omnibus Budget Reconciliation Act of 1993. This legislation created another benefit schedule for child-less tax filers, comprising a 7.65% subsidy over the first \$4000 of earnings and a maximum credit of \$306. Phase-in rates for families with children also increased substantially. Between 1994 and 1996, the phase-in rate for families with one child increased from 23.6% to 34%, while the rate for multiple-child families grew from 30% to 40%. By 1996, when the legislation was fully implemented, eligible families with children could receive maximum EITCs of \$2152 (for one child) or \$3556 (for multiple children).

3

Another important development is the proliferation of state EITCs. These programs "piggyback" onto the federal EITC by using its eligibility rules and calculating credits as a percentage of its rates. States have the option to structure their EITC programs as refundable or nonrefundable tax credits. Of the 17 states (including the District of Columbia) that implemented an EITC prior to 2005, 11 made the program refundable for the entire period of operation, while two states (Illinois and Maryland) changed from nonrefundable to refundable tax credits. Annual foregone revenue from state EITCs ranges from \$17 million in Vermont to \$591 million in New York (Nagle and Johnson, 2006).

Eligibility for the EITC is currently determined along several dimensions. Taxpayers must have non-zero earned income from wages or salary, business self-employment, or farm self-employment. In addition, single and married tax filers are eligible to claim the credit as long as adjusted gross income is below some threshold. This break-even point varies by year and the presence and number of children.<sup>3</sup> The EITC's original design extended eligibility only to families with children, but benefits did not vary with the number of children. However, as previously mentioned, legislative activity in 1993 created a small EITC for childless tax filers, and the 1990 legislation created a large and growing differential benefit for families with two or more children. With the exception of Wisconsin, all states utilize the federal credit's basic framework for the treatment of children. Wisconsin is the only state to include an incremental benefit for the third child. In particular, it supplements the federal credit by 4% for families with one child, 14% for families with two children, and 43% for families with three or more children.

A necessary condition for the EITC to alter fertility decisions is widespread awareness of and participation in the program. Scattered evidence indicates that this condition is likely to hold. Nearly two-thirds of parents with income below 200% of the federal poverty line (FPL) and over half of parents below 50% of the FPL are aware of the EITC (Phillips, 2001). Although much of this awareness comes from the growing number of public education campaigns launched by government agencies and local non-profits, low-income families increasingly use professional tax preparers to assist with claiming the credit. Indeed, ethnographic research by Romich and Weisner (2000) find that three-quarters of the families in their sample use either commercial or non-profit tax preparation services. Currently, about 62% of all EITC returns are filed with the help with a paid preparer, compared to 53% among non-EITC filers (IRS, 2000). Such broad awareness has translated into comparatively high participation rates. The best evidence to date comes from Scholz (1994), who estimates take-up rates between 80% and 86% as of the early-1990s, numbers that far exceed participation in AFDC/TANF (about 50%).

## 2.2. Theoretical framework

Standard economic models begin with the observation that fertility decisions are made in a series of steps, beginning with the decision about whether to have sex. Decisions are then made with respect to contraceptive intensity. If a pregnancy occurs, women must decide between aborting the pregnancy and giving birth. Two key assumptions underlie the standard model. Women are assumed to act with perfect information throughout this decision-making process, and they maintain perfect control over fertility outcomes. Based on these considerations, the model predicts that decreasing the costs associated with bearing and raising children reduces the likelihood that a pregnancy will end with an abortion. To the extent that the EITC supplements the wages of low-income families, this prediction suggests that the tax credit may lead to a decline in the abortion rate.

Some of these (relatively strong) assumptions can be relaxed by extending the model to account for the possibility that the EITC can influence abortion behavior at two points in the fertility decision-making process (Akerlof et al., 1996; Kane and Staiger, 1996; Levine, 2004).<sup>4</sup> First, the EITC can influence behavior when decisions are made about the level of sexual activity and contraceptive intensity. As others note, the EITC acts as insurance against income shocks by smoothing consumption when other resources are depleted (Kniesner and Ziliak, 2002). Women might therefore respond to this insurance by adopting behaviors that increase the risk of pregnancy. The central issue is determining how EITC-induced pregnancies would be resolved. For some women, the decrease in the cost of bearing and raising children would encourage a shift toward greater births. Once pregnant, other women might become increasingly aware of the logistical, financial, and emotional supports and challenges associated with obtaining an abortion (Levine, 2004). For example, some women may learn that abortion providers are highly accessible in the state of residence (leading to an increase in the number of abortions), while others could receive signals that the father is willing to support the child (leading to a reduction in the number of abortions). Overall, by encouraging women to become pregnant, the EITC can *increase* the number of abortions, although the net effect depends on how these additional pregnancies are treated.

Second, the EITC can alter abortion rates by changing women's decision-making after a pregnancy occurs. Since the credit acts as a wage supplement, an income effect is created that increases the demand for children, assuming that parents view them as normal goods. Therefore, the EITC is expected to reduce the number of abortions and increase the number of births. Also at work are price effects that operate through the EITC's differential benefit for families with different numbers of children. Prior to 1990, benefits did not vary with the number of children, and so the EITC created an incentive to reduce abortions only among childless women. The 1990 introduction of a separate benefit schedule for multiple-child families

<sup>&</sup>lt;sup>3</sup> The qualifying child must be a child, grandchild, stepchild, or foster child of the taxpayer, under 19 years old (under 24 if a full-time student), or permanently disabled, and lives with the taxpayer for the entire tax year.

<sup>&</sup>lt;sup>4</sup> These extensions come as a result of relaxing the assumption that women act with perfect information. Specifically, it is important to allow for the possibility that new information might be gained at several points in the decision-making process. For example, new information can arrive in the form of financial and emotional support from the father and other family members, as well as social attitudes toward non-marital childbearing.

4

#### C.M. Herbst/Social Science Research xxx (2011) xxx-xxx

created an additional price incentive to reduce abortions among women with one child. Women with two or more children do not experience a positive price incentive from the current EITC structure.

This simple model is complicated by the labor supply incentives introduced by the EITC. In particular, these incentives have implications for fertility decisions that differ according to whether a tax unit's earnings are in the phase-in, plateau, or phase-out region of the EITC. If earnings fall in the phase-in region, the credit acts as a wage supplement, thereby increasing the opportunity costs associated with bearing and raising children. On the other hand, the additional income from the EITC might encourage childbearing assuming that children are normal goods. Overall, the EITC creates ambiguous fertility incentives for women in the phase-in region. If earnings fall in the plateau region, where the credit operates like a lump sum transfer, the EITC creates ambiguous labor supply incentives. However, families experience an income effect that encourages a reduction in the number of abortions and an increase in the number of births. Finally, the EITC's implicit tax on earnings is predicted to reduce the number of abortions. In sum, regions of the EITC that create positive employment incentives create conflicting fertility incentives. Alternatively, regions of the credit that create ambiguous or negative employment incentives generally encourage childbearing through a pure income effect.

## 3. Literature review

Several strands of the empirical literature on births and abortions are relevant to the current study. In particular, a large body of research examines the impact of other means-tested programs, including AFDC/TANF benefits and recent reforms to the welfare system. In addition, a number of studies explore relationship between various tax policies (including the EITC) and fertility decisions. The discussion herein provides only a brief summary of each strand.

A review of research on AFDC/TANF benefits by Moffitt (1998) concludes that welfare generosity is likely to have a small positive and statistically significant effect on fertility. More recent work in this area focuses on the impact of welfare reform. There appears to be a convergence of evidence suggesting that welfare reform reduced fertility, particularly among teenagers. For example, Lopoo and DeLeire (2006) find that PRWORA's minor parent provisions led to a relative decline in fertility rates of 0.7% points among 15–17-year-olds. This result is corroborated by Kaestner et al. (2003) and Offner (2003), both of which find that the implementation of PRWORA led to small but statistically significant reductions in teenage childbearing.

Also relevant to the current study are evaluations of family cap policies, which deny families incremental increases in welfare benefits as the number of children increases. Studies of family caps using experimental designs have been conducted for two states. Turturro et al. (1997) finds no impact of family caps on births in Arkansas, and Camasso et al. (1999) finds that New Jersey's family cap reduced the number of pregnancies and births and increased the number of abortions. A number of studies based on observational data have been conducted as well, also producing mixed results. Horvath-Rose and Peters (2001) find that family caps are associated with lower fertility rates, while Joyce et al. (2004) detect no effects. Interestingly, recent work by Levine (2002) and Kearney (2004) find small (but largely insignificant) positive effects of family caps on births.

The final strand of empirical research focuses on the impact of the income tax code. Whittington et al. (1990) use aggregate time series data between 1913 and 1984 to examine the influence of the personal exemption. These authors determine that increases in the personal exemption are positively related to childbearing, a result confirmed Whittington (1992), who uses microdata between 1979 and 1983 to examine married couples. More recent work by Dickert-Conlin and Chandra (1999) investigates the role of taxes in the timing of fertility. Using a sample of children born between late-December and early-January, the authors find that a \$500 increase in the tax benefit associated with a December birth increases the likelihood of such a birth by 17% points.

Only two studies examine the impact of the EITC on births. The first, by Duchovny (2001), uses a difference-in-differences approach to evaluate the 1990 tax legislation that introduced an incremental increase in EITC benefits for multiple-child families. Using as the comparison group families with two or more children, the author finds that the tax reform increased fertility only among married white women. The second study, by Baughman and Dickert-Conlin (2007), draws upon aggregate natality data between 1990 and 1999 from the National Center for Health Statistics. These authors estimate birth rate models that include the EITC maximum benefit and the incremental benefit. Interestingly, results suggest that increases in both are associated with *reductions* in first and higher-order births, especially among white women.

## 4. Empirical approach

## 4.1. Primary data source: CDC counts of state-level abortion occurrences

Abortion data for this research are drawn from the CDC's annual publication *Abortion Surveillance*.<sup>5</sup> A data collection initiative that began in 1969, the CDC receives on a voluntary basis state-level abortion counts from public health agencies, hospitals, and other medical facilities. Many of the public health agencies obtain information directly from abortion providers. The CDC abortion counts reflect only legally-induced abortions, defined as those performed by a licensed physician or an individual

<sup>&</sup>lt;sup>5</sup> Detailed information on these reports, as well as the abortion data themselves, can be found at http://www.cdc.gov/reproductivehealth/Data\_Stats/Abortion.htm.

Please cite this article in press as: Herbst, C.M. The Earned Income Tax Credit and abortion. Social Sci. Res. (2011), doi:10.1016/ j.ssresearch.2011.05.003

acting under the supervision of a physician. I use this information to construct a state-by-year dataset of abortions between 1975 and 2005.<sup>6</sup> The CDC abortion counts are available for all state-year combinations except in Alaska, California, Louisiana, New Hampshire, Oklahoma, and West Virginia.<sup>7</sup> These missing data reduce the number of state-year combinations by 26, leaving a total of 1555 observations in the analysis.

These data contain a number of noteworthy characteristics. Perhaps the most important is that, until recently, states' public health agencies reported data according to the state in which abortions were performed, rather than the state in which women resided. This artificially increases the abortion rate in states with large numbers of out-of-state residents crossing the border to obtain an abortion.<sup>8</sup> States vary considerably in the degree to which non-residents use in-state abortion services. As of 2005, for example, about half of abortions in the District of Columbia were performed on out-of-state residents, whereas none of the abortions in Wyoming were performed on such residents. Another issue deals with the well-known undercount of CDC's abortion occurrences. Analyses by Blank et al. (1996) find that CDC counts are consistently below those of the Alan Guttmacher Institute (AGI), another key source of aggregate abortion data in the US.<sup>9</sup> Indeed, I find that in the period covered in the current analysis, annual AGI abortion counts exceed the CDC counts by approximately 22%, on average.

## 4.2. Empirical model

Using state-level data on abortion occurrences between 1975 and 2005, I estimate permutations of the following OLS regression model:

$$\mathbf{y}_{st} = E_{st}\boldsymbol{\beta} + \mathbf{P}_{st}^{\prime}\boldsymbol{\psi} + \mathbf{D}_{st}\boldsymbol{\gamma} + \boldsymbol{\mu}_{s} + \boldsymbol{\upsilon}_{t} + \boldsymbol{\varepsilon}_{st},\tag{1}$$

for s = 1, ..., S; t = 1, ..., N, where *s* indexes states and *t* indexes years. The dependent variable, *y*, represents one of several measures of aggregate abortion occurrences. I rely primarily on the abortion rate, defined as the number of abortions within each state-year cell divided by the number of women ages 15–44 (and multiplied by 1000). This variable should be interpreted as the number of abortions per 1000 women of childbearing age. In specification checks, I estimate models in which the dependent variable is the number of abortions, adding the state's female population ages 15–44 as a control. The spread of abortions across states and over time is not normally distributed, so all outcome variables are expressed in logarithmic form.<sup>10</sup>

The key right-hand-side variable in (1) is denoted by E, the combined federal and state EITC maximum credit.<sup>11</sup> Although E is defined as the contemporaneous EITC, I experiment with models that include a one-period lag of the credit. This lag structure accounts for the gap that exists between the time fertility decisions are influenced by the EITC and when those decisions manifest in the data. By parameterizing E as the maximum credit, I capture the income effect associated with the introduction and subsequent expansion of the EITC.

I rely on several sources of geographic (cross-state) and temporal (over time) variation to identify the impact of the EITC. The study period includes three major federal expansions (in 1986, 1990, and 1993), each one substantially increasing the maximum credit. The largest of these expansions came in 1993, when the phase-in rate increased from 18.5% to 34% for families with one child and from 19.5% to 40% for families with multiple children. Furthermore, the 1990 reform created the incremental benefit for families with two or more children. The federal credit was adjusted for inflation in 1986, approximately 10 years after its implementation, providing another source of policy variation during the observation period. There is also substantial cross-state variation in EITC benefits. As of 2005, 16 states (and the District of Columbia) operated their own EITC program, thus becoming an important mechanism for offsetting state income taxes.<sup>12</sup> Previous studies rely on this

<sup>&</sup>lt;sup>6</sup> Between 1970 and 1974, only eight to 37 state public health agencies reported abortion information, although many hospitals contributed to the data collection. Beginning in 1975, 41 agencies reported this information, reaching a peak of 49 agencies during the period 2000 to 2005. Given data quality concerns in the years immediately after *Roe v. Wade*, I follow other researchers by omitting 1973 and 1974 from the analysis. Therefore, the first year of my observation period coincides with the introduction of the EITC.

<sup>&</sup>lt;sup>7</sup> Abortion data are missing for Alaska between 1998 and 2002, for California between 1998 and 2005, for Louisiana in 2005, for New Hampshire between 1998 and 2005, for Oklahoma during 1998 and 1999, and for West Virginia during 2003 and 2004.

<sup>&</sup>lt;sup>8</sup> This also creates an inconsistency in the abortion rate measure: whereas as the numerator commingles abortions to in-state and out-of-state residents, the denominator includes only in-state residents of childbearing age (ages 15-44) (Blank et al., 1996).

<sup>&</sup>lt;sup>9</sup> The AGI collects abortion data in a different manner. Specifically, it conducts periodic surveys of abortion providers, and is known to maintain an extensive list of such providers. Although the AGI is widely regarded as providing the most accurate abortion data, it too is not without its drawbacks. First, unlike the CDC reports, the AGI survey is not conducted annually, and has become less frequent over time. During the period 1975–2005, AGI abortion data are only available for the following years: 1975–1982, 1984–1985, 1987–1988, 1991–1992, 1995–1996, 2000, and 2005. The increased data gaps after 1990 are especially problematic, given that the EITC experienced its largest expansions during this time. In addition, analyses by Michael (1999) and Joyce and Kaestner (1996) reveal somewhat contradictory but nevertheless problematic characteristics of AGI data. Michael (1999) finds that AGI may in fact overstate the number of abortions, while Joyce and Kaestner (1996) argue that overall reporting of AGI abortion counts has declined over time. Finally, AGI data do not disaggregate abortion counts by the number of previous live births or demographic sub-group.

<sup>&</sup>lt;sup>10</sup> For 2000 and 2001, Wyoming reports zero abortions among women with one previous live birth. To construct the logged abortion rate for this group, I omit these two observations from the analysis. I also experimented with imputing a value of one abortion for these state-year cells, and the results are not sensitive to this imputation.

<sup>&</sup>lt;sup>11</sup> I draw from several sources to construct the EITC variables. Federal EITC parameters are from the *Green Book* (US House of Representatives, 2004) and various publications from the Center on Budget and Policy Priorities. The state parameters are also taken from the Center on Budget and Policy Priorities, as well as Fang and Keane (2004) and Dickert-Conlin & Houser (2002).

 $<sup>^{12}</sup>$  In 1999, Colorado implemented an EITC, but it was suspended in 2002 because of insufficient funds.

6

#### C.M. Herbst/Social Science Research xxx (2011) xxx-xxx

federal and state variation to identify the credit's impact on employment (Grogger, 2003; Herbst, 2008), welfare participation (Fang and Keane, 2004; Looney, 2005) and fertility (Baughman and Dickert-Conlin, 2007). The key assumption in using the combined federal and state maximum credit is that all legislative changes are exogenous to fertility decisions. This assumption is plausible, especially for the federal component of the EITC, since tax units in all states are exposed to the same eligibility rules and benefit schedules.<sup>13</sup>

Eq. (1) incorporates a large vector of additional policy controls (**P**'). Specifically, I include three variables measuring restrictions on abortion access: federal and state bans on the use of Medicaid funds to cover abortions, parental notification or consent laws for minors seeking an abortion, and mandatory waiting periods before an abortion can be performed.<sup>14</sup> I follow previous researchers by coding each abortion restriction according to the fraction of the year it was in effect (Bitler and Zavodny, 2001; Blank et al., 1996). This is particularly important for the Medicaid funding restriction, given that its implementation was delayed and enjoined in the years following the 1976 Congressional ban on the use of federal funds to pay for abortions (Hyde Amendment).<sup>15</sup> To account for the possibility that abortions in a given state are influenced not only by its own restriction policies but also those in nearby states, I incorporate dummy variables that equal unity if a border state enacts Medicaid funding, parental involvement, and mandatory delay laws. Inclusion of these variables also helps to mitigate the bias that may result from using abortion data based on state of occurrence.

Also included in **P**' is a number of social policy controls. I account for the flurry of welfare reform legislation implemented throughout the 1990s by including dummy variables for welfare waivers and the Personal Responsibility and Work Opportunity Reconciliation Act (PRWORA) of 1996. Only statewide welfare waivers for work requirements, benefit termination time limits, work requirement time limits, and sanctions are considered in this analysis. I code both variables as equal to the fraction of the year in which the policy is "turned on." I also control for the generosity of states' welfare benefits through the maximum monthly AFDC/TANF benefit paid to non-working recipients (in a three-person family). Finally, I incorporate a variable capturing the fraction of the year in which a family cap is implemented through welfare waivers or PRWORA. Inclusion of these social policy controls is important because they alter the costs and benefits of participating in a cash assistance program (AFDC/TANF) that conditions eligibility on having one or more children.<sup>16</sup>

The final set of observables included in (1) focuses on states' economic, political, and demographic environments ( $\mathbf{D}'$ ). Changes in fertility behavior due to economic conditions are controlled for by the average, annual state unemployment rate. Differences over time and across states in the political and policy-making environments are accounted for through a dummy variable that equals unity for state-years in which there is a Republican governor. Also included is a continuous variable measuring the fraction of each state's population voting Republican in the previous Presidential election. Finally, I incorporate several demographic and human capital characteristics that may shift preferences for bearing and raising children. These variables include population density, the overall marriage rate, the fraction of the population ages 25 and over with at least a bachelor's degree, and indicators of the female age structure.

As previously stated, an advantage of the panel data used in this study is the ability to control for many unobserved attributes of states, which may simultaneously influence EITC policymaking and abortion decisions. One concern is that states implementing a tax credit could be systematically different from those that do not in terms of political and cultural attitudes toward abortion. If the processes leading to a state EITC are not properly accounted for, coefficients on the EITC will be biased. Therefore, state fixed effects,  $\mu_s$ , are added to control for unobserved time-invariant differences between states, including differences in abortion attitudes and preferences for certain types of social policies. I also add year dummy variables,  $v_t$ , to filter out year-to-year changes in fertility behavior that are common to all states, driven for example by nation-wide economic shocks or downturns, Supreme Court decisions related to abortion policy, and federal social policy reforms (e.g., welfare reform and the child care tax credit). Finally, I experiment with models that add state-specific time trends, which control for unobserved factors that trend linearly within states over the observation period. All regressions are

<sup>&</sup>lt;sup>13</sup> The potential endogeneity of the EITC is a manifestation of the more general policy endogeneity problem. The concern here is that the timing of EITC expansions coincides with, or responds to, broader societal trends that influence abortion propensities. For example, if states that enact EITC programs to reduce poverty or strengthen work incentives have populations with different underlying views on abortion, then EITC policies will be correlated with these unobserved preferences and the estimated effect of the credit on abortion will be biased. However, as noted in the text, this is not likely to be a major problem in the current study. Given that most of the variation in the EITC comes from federal policy changes—and hence all individuals, regardless of the state of residence and attitudes toward abortion, are exposed to the same benefit rules—it is not likely that EITC policymaking is related to state-specific policy environments or cultural shifts related to abortion. Nevertheless, if state legislators use the EITC to respond to changes in employment and poverty or gain electoral advantage by providing tax relief to low-income individuals, then one might expect future changes in the EITC to be correlated with current abortion rates. Therefore, I conduct specification checks that include a one-year lead of a state EITC dummy variable in the abortion regression. In no case do I find that future state policy changes are related to current abortion rates.

<sup>&</sup>lt;sup>14</sup> Data for these variables are drawn from a number of sources, including Levine (2002; 2003; 2004), Merz et al. (1995), Sollom (various years), and the National Abortion Rights Action League (NARAL) annual publication *Who Decides*?. I am also grateful to Rebecca London for generously sharing the abortion restriction codes used in Blank et al. (1996).

<sup>&</sup>lt;sup>15</sup> Note that my coding does not make a distinction between periods during which abortion restrictions are present but nevertheless enjoined and periods during which such laws do not exist. In this sense, my coding is similar to that of Joyce and Kaestner (2001), and Levine (2002; 2003). A number of studies create separate variables for enjoined abortion restrictions. However, they apply to a very small number of state-year combinations and either have not been found to influence abortion decisions or do so in a way that conflicts with theoretical predictions (Bitler and Zavodny, 2001; Blank et al., 1996).

<sup>&</sup>lt;sup>16</sup> Data on the timing and nature welfare waiver and TANF reforms are drawn from Crouse (1999), US DHHS (1997), and US GAO (1997). Maximum welfare benefits are drawn from the *Green Book* (US House of Representatives, 2004) and the Urban Institute's *Welfare Rules Databook* (Rowe and Versteeg, 2005). To code states' family cap policies, I use the information provided in Kearney (2004) and the *Welfare Rules Databook* (Rowe and Versteeg, 2005).

## Table 1

Variable	Definition	Mean	SD
Abortion rate	Number of abortions per 1000 women ages 15–44, based on CDC data	20.94	11.05
Maximum EITC	Combined maximum federal/state EITC for families with two or more children (\$1000)	1.967	1.633
Federal EITC	Federal maximum EITC for families with two or more children (\$1000)	1.900	1.538
State EITC	State EITC, based on state-specific and federal credit rates (\$1000)	0.595	0.384
Medicaid	=1 if a state implements a restriction on using Medicaid funds to pay for abortions	0.556	0.489
Involve	=1 if a state implements a parental notification or consent law for minors seeking an abortion	0.316	0.465
Delay	=1 if a state implements a mandatory delay law	0.123	0.328
Medicaid border	=1 if a border state implements a restriction on using Medicaid funds to pay for abortions	0.871	0.335
Involve border	=1 if a border state implements a parental notification or consent law for minors seeking an abortion	0.692	0.461
Delay border	=1 if a border state implements a mandatory delay law	0.328	0.469
Welfare waiver	=1 if a state implements a statewide welfare waiver	0.057	0.219
TANF	=1 if a state implements TANF	0.277	0.442
Welfare benefit	Maximum monthly AFDC/TANF benefit for a three-person family (\$1000)	0.358	0.145
Family cap	=1 if a state implements a family cap on incremental increases in welfare benefits	0.144	0.347
UR	Average annual state unemployment rate (/100)	0.063	0.019
Governor	=1 if the current state governor is Republican	0.496	0.500
President	Percent Republican vote in the previous Presidential election	0.493	0.092
Population density	Number of individuals per square mile (/100)	2.469	5.704
Marriage	Marriage rate, per 1000 in the population (/100)	0.094	0.070
BA	Percent of population ages 15 and over with at least a bachelor's degree	0.207	0.053
Females ages 15–19	Percent of female population ages 15–44 between ages 15 and 19	0.166	0.024
Females ages 15–44	Female population ages 15–44 (/1000)	2321.660	1777.079

*Notes*: Summary statistics in this table are derived from state panel data between 1975 and 2005 in which state-year combinations of CDC abortion data are non-missing (unbalanced panel N = 1555). There are two state-year combinations with zero one-child abortions (Wyoming, 2000 and 2001); these are dropped from the regressions, given that the log of all abortion rates is taken. Summary statistics for the State EITC are based those state-years with a state EITC implemented. All summary statistics are weighted by the female population ages 15–44.

weighted by the female population ages 15–44, and standard errors are corrected for arbitrary heteroskedasticity using robust standard errors.

## 5. Results

## 5.1. Summary statistics

Table 1 presents summary statistics for the key abortion variables in this analysis. The overall abortion rate is approximately 21 abortions per 1000 women of childbearing age. Stated another way, 2.1% of such women have an abortion in a given year. These figures mask substantial year-to-year variation in abortion rates: after peaking at about 24 abortions per 1000 women throughout the early-1980s, the abortion rate declined to a low of 16 between 2000 and 2005. Indeed, a regression of the (log) abortion rate on a linear time trend suggests that the abortion rate between 1975 and 2005 declined about 1.5% per year. Table 1 also shows that the standard deviation around the mean abortion rate (for the entire study period) is 11.1 abortions per 1000 women of childbearing age. Within-state, over time variation in abortion rates is comparatively low, ranging from 1.4 standard deviations in New Mexico to 7.1 standard deviations in California.<sup>17</sup> Thus, most of the variation in abortion rates comes from cross-state differences.<sup>18</sup>

Descriptive results for the EITC and other policy variables are also provided in Table 1. The real value of EITC maximum credit for families with one child increased about 94% between 1975 and 2005, while the value of the credit for multiplechild families increased over 200%. Given that the impact of the EITC is identified using primarily within-state variation, Table 1 disaggregates the combined maximum credit into its federal and state components. Among states with an EITC, there is sizeable variation in benefits (*SD* = \$384) around the mean benefit (\$595).<sup>19</sup> A Medicaid restriction was in effect for 56% of the study period, parental involvement laws were in place for 32% of period, and mandatory delay legislation covered only 12% of the sample period. Table 1 reveals that border state policies are potentially important influences on fertility behavior: fully 70% of adjacent states implemented a parental involvement law between 1975 and 2005, and another 33% of adjacent states enacted mandatory waiting periods. Finally, welfare reform legislation through AFDC waivers was in place for 6% of the study period, and PRWORA was in effect for approximately 28% of the study period.

<sup>&</sup>lt;sup>17</sup> The District of Columbia is a clear outlier in these data, with an average abortion rate of 109 per 1000 women of childbearing age and a standard deviation of 45. Robustness checks with the District omitted leave the main results intact.

 $<sup>^{18}</sup>$  Indeed, a regression of the abortion rate on state and year fixed effects produces an  $R^2$  of 83%, suggesting that only 17% of the variation in abortion rates is due to factors that change within states over time.

<sup>&</sup>lt;sup>19</sup> Another way to measure the amount of within-state variation in the EITC is to count the number of times states altered the credit rate between 1975 and 2005. Across all states, the average state changed its credit rate 1.02 times (the minimum number of policy changes is zero and the maximum is eight). Among states with an EITC, the average number of changes is approximately three. This suggests a modest amount of within-state policy variation, although it should be considered a lower bound, as the actual state credit *amount* changed each year due to indexing for inflation.

#### 5.2. Estimation results

Main results from the abortion regressions are shown in Table 2. It presents estimates from OLS regressions of the overall abortion rate on the EITC maximum credit. Columns (1)–(3) show results from the baseline model, and columns (4)–(7) present a number of specification checks. Specifically, I test the sensitivity of the basic results to the inclusion of state-specific time trends, the log number of abortions as the dependent variable, a one-period lag of the EITC maximum credit.

The estimate in column (1) of Table 2 suggests that a \$1000 increase in the combined EITC maximum credit is associated with a decrease in the abortion rate of 8%. However, this model does not account for observable state factors, as well as unobserved heterogeneity, that may influence individual fertility and state-level social policy decisions. When the full set of policy, economic, demographic, and political controls are added in column (2), the estimated effect of the EITC increases dramatically. The coefficient implies that a \$1000 increase in the EITC is associated with a 17% decrease in the abortion rate. Adding state fixed effects and year dummies [column (3)] reduces the impact of the EITC to a 7.6% reduction in the abortion rate. The coefficient remains statistically significant at conventional levels. This effect implies a reduction of approximately 1.6 abortions per 1000 women of childbearing age.

Results presented in columns (4)–(7) suggest that the EITC effect is fairly robust to a number of specification checks. Including state-specific time trends further reduces the magnitude of the EITC coefficient, rendering it imprecisely estimated. However, substituting the log number of abortions for the log abortion rate produces a statistically significant coefficient on the EITC.<sup>20</sup> The estimate implies that a \$1000 increase in the maximum EITC leads to a 7.7% reduction in the number of abortions. Columns (6) and (7) explore the impact of a one-period lag in the maximum credit. Column (6) uses as the dependent variable the log abortion rate, and column (7) uses the log number of abortions. State-specific time trends are included in both models. Parameter estimates reveal that the impact of the EITC is robust to changes in the lag structure. If anything, the lagged results imply slightly larger effects, especially in the model using the log number of abortions. Results from this model show an 11% reduction in the number of abortions in response to a \$1000 increase in the EITC's maximum credit.<sup>21</sup>

Turning to the remaining policy variables, I find mixed evidence for the impact of abortion restrictions and other policy reforms. Bans on Medicaid funding are associated with reductions in abortion, but the effect becomes statistically insignificant once state-specific time trends are added to the model. On the other hand, the impact of mandatory delay laws doubles in magnitude and becomes statistically significant once the time trends are included. Consistent with previous studies, border state policies have conflicting and counter-theoretical effects on abortion. For example, border state restrictions on teen abortions are associated with decreased in-state abortion rates, a result contrary to theoretical predictions but nevertheless mirrored by other studies (Blank et al., 1996; Bitler and Zavodny, 2001). Finally, the results provide mixed evidence on the impact of social policy reforms. The negative coefficient on welfare waivers is inconsistent with a theoretical model of welfare reform (Klerman, 1998). Although family caps are predicted to increase abortion rates, the policy is associated with reductions in abortion. This result is consistent with previous work by Kearney (2004) and Levine (2003), who find that family caps lead to small but imprecisely estimated increases in birth rates.

## 5.3. Births and pregnancies

The results in Table 2 suggest that expansions to the EITC are associated with reductions in the overall abortion rate. As the theoretical framework makes clear, the EITC has implications for abortion decisions at two points in the fertility decisionmaking process: when choices are made regarding sexual and contraceptive practices and after a pregnancy occurs. If the EITC does not alter sexual and contraceptive behavior, a reduction in the number of abortions should be attributable to an increase in the number of births. On the other hand, if the EITC influences the intensity with which women use contraception, the reduction in abortion rates can be traced to a decline in pregnancies.

The results presented in Table 3 allow us to examine these pathways. Estimated in columns (1) and (2) are OLS regressions of state-level birth rates on a one-period lag of the EITC maximum credit. The mean birth rate between 1975 and 2005 is about 67 births per 1000 women of childbearing age. Columns (3) and (4) show results from OLS regressions of pregnancy rates on the identical set of variables. Following Levine (2002, 2003), I construct the pregnancy rate by summing the birth and abortion rates.<sup>22</sup> The mean pregnancy rate over the study period is 88 pregnancies per 1000 women of childbearing age. The odd-numbered columns estimate models without state-specific time trends, and the even-numbered columns add these controls to the model. Although birth rate data are available for all state-years during the study period, for consistency I limit

<sup>&</sup>lt;sup>20</sup> Defining the outcome variable in terms of the log number of abortions is tantamount to moving the denominator of the abortion rate (number of women ages 15 to 44) from the left-hand-side of the equation into the right-hand-side. Therefore, in all models that include the log number of abortions, I add a control for the log number women ages 15 to 44.

<sup>&</sup>lt;sup>21</sup> In a further robustness check, I estimate a model in which all of the right-hand-side variables are lagged one year. Doing so does not change the estimated effects of the EITC on abortion rates.

<sup>&</sup>lt;sup>22</sup> Creating a pregnancy rate based on birth and abortion rates necessarily misses pregnancies that end in a spontaneous abortion (i.e., a miscarriage). Therefore, the pregnancy rate used in this analysis is likely to be below the actual rate. Determining the rate at which pregnancies are spontaneously aborted, however, is a difficult task because many miscarriages occur very early in a pregnancy, often before a woman realizes she is pregnant. Estimates by Wilcox et al. (1999) and Wang et al. (2003) suggest that 25% of pregnancies are miscarried by the 6th week, while clinical miscarriages—those occurring after the 6th week—occur in about 8% of pregnancies (Wang et al., 2003).

Maximum EITC         -0.080*** (0.011)         -0.172*** (0.028)           Maximum EITC <sub>t-1</sub> -0.135*** (0.030)         0.030)           Medicaid         -0.240*** (0.030)           Involve         -0.240*** (0.031)					
	$-0.076^{**}$ (0.035)	$-0.055\ (0.048)$	$-0.077^{st} \left( 0.046  ight)$	0,000* (0,045)	
	0.068*** (0.033)		0,000,000	(CFU) 080.0- (CCUA) 9000	-0.112 (0.043)
	0.001 (0.021)	-0.025 (0.018)	-0.020(0.022)	-0.020 (0.022) 0.025 (0.018)	-0.020 (0.022) 0.023 (0.017)
	-0.022 (0.019)	$-0.043^{**}(0.020)$	$-0.050^{**}(0.020)$	-0.044** (0.020)	$-0.052^{**}(0.020)$
Medicaid border –0.053 (0.035)	$0.064^{*}$ (0.037)	0.052 (0.043)	0.047 (0.042)	0.055(0.043)	0.050 (0.042)
Involve border –0.125*** (0.035)	-0.023 (0.020)	$-0.044^{**}(0.020)$	$-0.044^{**}$ (0.020)	$-0.045^{**}$ (0.021)	$-0.044^{**}$ (0.020)
Delay border –0.026 (0.048)	0.001 (0.020)	0.017 (0.017)	0.006 (0.017)	0.017 (0.017)	0.005 (0.017)
Welfare waiver 0.056 (0.067)	0.012 (0.023)	$-0.042^{**}$ (0.021)	$-0.044^{**}$ (0.021)	$-0.040^{st}$ (0.020)	$-0.041^{*}(0.021)$
TANF 0.258*** (0.069)	0.013(0.043)	0.049 ( $0.041$ )	0.027 (0.044)	0.051(0.042)	0.027 (0.045)
Welfare benefit 0.114)	0.169(0.166)	0.266(0.162)	0.194(0.168)	0.261(0.161)	0.185(0.166)
Family cap -0.065 (0.046)	-0.030(0.020)	-0.029 (0.025)	-0.031(0.025)	-0.030(0.025)	-0.033(0.025)
DV: In(abortion rate) 🗸	>	>		>	
DV: In(No. of abortions)			>		>
State characteristics No Yes	Yes	Yes	Yes	Yes	Yes
State fixed effects No No	Yes	Yes	Yes	Yes	Yes
Year dummies No No	Yes	Yes	Yes	Yes	Yes
State-specific time trends No No	No	Yes	Yes	Yes	Yes
R <sup>2</sup> 0.066 0.493	0.877	0.928	0.987	0.928	0.987
State-year combinations 1555 1555	1555	1555	1555	1555	1555

Table 2

C.M. Herbst/Social Science Research xxx (2011) xxx-xxx

9

#### 10

#### C.M. Herbst/Social Science Research xxx (2011) xxx-xxx

## Table 3

The impact	of the	EITC on	births and	d pregnancies.

Variable	(1)	(2)	(3)	(4)
Maximum EITC <sub>t-1</sub>	$-0.037^{***}$ (0.010)	$-0.082^{***}$ (0.029)	$-0.063^{***}$ (0.009)	$-0.068^{***}$ (0.018)
Medicaid	0.004 (0.008)	-0.001 (0.009)	$-0.012^{*}(0.006)$	-0.010 (0.006)
Involve	$-0.025^{***}$ (0.004)	$-0.010^{***}$ (0.003)	$-0.025^{***}$ (0.004)	-0.011**** (0.003)
Delay	$-0.028^{***}$ (0.005)	0.013** (0.006)	$-0.018^{***}$ (0.005)	0.003 (0.006)
Medicaid border	$-0.021^{**}$ (0.009)	0.013 (0.021)	0.006 (0.009)	0.022 (0.013)
Involve border	-0.012 (0.009)	0.007 (0.007)	$-0.023^{***}$ (0.007)	-0.000(0.006)
Delay border	0.013** (0.005)	0.008 (0.006)	0.015*** (0.006)	$0.012^{*}(0.006)$
Welfare waiver	0.031**** (0.010)	-0.000 (0.007)	0.026*** (0.011)	$-0.015^{*}(0.008)$
TANF	-0.009(0.019)	0.042 (0.031)	-0.000(0.022)	0.048 (0.036)
Welfare benefit	0.428**** (0.065)	0.266**** (0.055)	0.410**** (0.065)	0.357*** (0.066)
Family cap	0.017**** (0.006)	0.018* (0.009)	0.008 (0.006)	0.017** (0.008)
DV: ln(birth rate)	$\checkmark$	$\checkmark$		
DV: ln(pregnancy rate)			$\checkmark$	$\checkmark$
Mean of birth rate	66.51			
Mean of pregnancy rate			87.46	
State characteristics	Yes	Yes	Yes	Yes
State fixed effects	Yes	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes	Yes
State-specific time trends	No	Yes	No	Yes
$R^2$	0.834	0.906	0.900	0.948
State-year combinations	1.555	1.555	1.555	1.555

*Notes*: Regression coefficients in this table come from the unbalanced panel of state-year combinations between 1975 and 2005, and are weighted by the female population ages 15–44. Although fertility data are available for all state-years, for consistency the analysis is restricted to state-years with non-missing CDC abortion data. Standard errors, reported in parentheses, are corrected for arbitrary heteroskedasticity using Huber–White robust standard errors. Models (1) and (2) and (6) use the log fertility rate as the dependent variable, while models (3) and (4) use the log pregnancy rate as the dependent variable. The pregnancy rate is constructed by summing the CDC abortion rate and the fertility rate (per 1000 women ages 15–44). All models parameterize the EITC variable by once-lagging the 2 + -child maximum credit. All models control for UR, governor, president, population density, marriage, BA, and females ages 15–19. See Table 1 for an explanation of the variable definitions.

<sup>\*\*\*</sup> Indicate statistical significance at the 0.01 levels.

\*\* Indicate statistical significance at the 0.05 levels.

\* Indicate statistical significance at the 0.10 levels.

the analysis to the observations for which abortion data are available (N = 1555). Both dependent variables are expressed in logarithms.

A number of interesting findings emerge from Table 3. Increases in the maximum EITC are associated with *reductions* in aggregate birth rates. As shown in column (2), a \$1000 increase in the maximum credit is expected to decrease birth rates by about 8%. Such results find some support in the empirical literature. For example, Baughman and Dickert-Conlin (2007) determine that EITC expansions are associated with decreases in first and higher-order births among white women. Moreover, recent work by Kearney (2004) finds that lowering incremental welfare benefits through family caps leads to small increases in birth rates. Based on the theoretical model, if the EITC is associated with reductions in birth and abortion rates, the credit must encourage women to alter sexual and contraceptive behavior, as reflected in pregnancy rates. Results in column (4) corroborate this: a \$1000 increase in the maximum credit is associated with a 7% decline in pregnancy rates. Therefore, it appears that the EITC changes fertility outcomes by altering earlier decisions about the level of sexual activity and contraceptive intensity.

## 5.4. Specification checks

The main results are subjected to a number of sensitivity tests, as shown in Tables 4 and 5. Table 4 examines the impact of the EITC across several demographic sub-groups. I also explore the sensitivity of the main results to changes in the measurement and functional form of the EITC variable, alternative weights and data restrictions, and a correction for autocorrelation. Table 5 compares results using CDC abortion counts with those drawn from the AGI. As others note, the impact of abortion restrictions is somewhat sensitive to the abortion data source, making such a comparison important for the EITC as well (Bitler and Zavodny, 2001; Levine, 2003; Matthews et al., 1997).

The first two rows in Table 4 explore heterogeneity in the impact of the EITC across sub-groups of women defined by race.<sup>23</sup> I use as the dependent variable the log number of abortions among women in the relevant sub-group. The coefficient on the maximum credit shows a statistically significant 20% reduction in the number of abortions for black women, but a smaller and imprecisely estimated reduction for white women. These findings are contrary to those reported in Duchovny (2001) and Baughman and Dickert-Conlin (2007), both of which find greater fertility responses among white women. In results not reported, I estimate models that substitute the incremental EITC for the maximum credit. Across all sub-groups, increases in

<sup>&</sup>lt;sup>23</sup> As previously noted, these analyses are constrained to the period 1990–2005. These sub-group analyses include state and year fixed effects, state-specific time trends, and a control for the log number of women ages 15–44.

### C.M. Herbst/Social Science Research xxx (2011) xxx-xxx

## Table 4

Specification checks.

	Coefficient on the EITC		
Specification	(1)	(2)	
DV: Log of abortions to black women	-	$-0.200^{***}$ (0.072)	
DV: Log of abortions to white women	-	-0.139 (0.093)	
Abortion rate and the number of abortions(no logs)	$-3.045^{***}$ (0.528)	-15,453*** (3524)	
Log of maximum EITC	$-0.221^{*}(0.130)$	-0.177 (0.133)	
Maximum EITC for families with one child	-0.088(0.058)	-0.062(0.061)	
Analysis of 1990–2005 data, refundable state EITCs	$-0.104^{***}$ (0.038)	$-0.113^{***}$ (0.038)	
Balanced panel data	$-0.090^{*}$ (0.046)	$-0.093^{**}$ (0.046)	
Unweighted	$-0.179^{***}$ (0.048)	$-0.165^{***}$ (0.047)	
Prais–Winsten correction for serial correlation, AR(1) process	$-0.088^{**}$ (0.040)	$-0.076^{*}$ (0.040)	
Prais-Winsten correction for serial correlation, AR(1) process, and panel-corrected standard errors	$-0.088^{*} (0.045)$	$-0.076^{*}$ (0.046)	
Exclude states with at least 20% of abortions performed on non-residents	$-0.095^{**}$ (0.037)	$-0.100^{**}$ (0.039)	
Exclude states with at least 10% of abortions performed on non-residents	$-0.084^{**}$ (0.037)	$-0.086^{**}(0.038)$	
Add control for whether border state has an EITC program	$-0.070^{*}$ (0.036)	-0.059(0.038)	
Include additional demographic controls	$-0.072^{*}$ (0.037)	$-0.064^{*}$ (0.038)	
DV: ln(abortion rate)	$\checkmark$		
DV: ln(No. of abortions)		$\checkmark$	

*Notes*: Unless otherwise noted, regression coefficients in this table come from the unbalanced panel of state-year combinations between 1975 and 2005, and are weighted by the female population ages 15–44. Standard errors, reported in parentheses, are corrected for arbitrary heteroskedasticity using Huber–White robust standard errors. In addition, unless otherwise noted, the coefficient presented is the estimate on Maximum EITC, the combined federal/ state EITC for families with two or more children. The dependent variable in the first two models is the log number of abortions black (*N* = 579), and white (*N* = 585) women. Abortion data by demographic group are available between 1990 and 2005. The number of observations in the balanced panel regressions is 1395. This model includes state-specific time trends. The Prais–Winsten models assume an AR(1) process that is common across all states. The model with additional demographic controls includes the percent of women in 5-year age-groupings between ages 15 and 44. All models control for UR, governor, president, population density, marriage, BA, and females ages 15–19, while the estimates in (2) control for the log female population ages 15–44. All models include state fixed effects and year dummies. Models one through four and model nine include state-specific time trends.

<sup>\*\*\*\*</sup> Indicate statistical significance at the 0.01 levels.

\*\* Indicate statistical significance at the 0.05 levels.

\* Indicate statistical significance at the 0.10 levels.

## Table 5

Comparison of empirical results using CDC and AGI abortion data.

Variable	(1a)	(1b)	(2a)	(2b)	(3a)	(3b)
Maximum EITC	$-0.070^{***}$ (0.017)	$-0.075^{***}$ (0.016)	$-0.242^{***}$ (0.043)	-0.267*** (0.041)	0.002 (0.067)	0.036 (0.048)
Medicaid			$-0.157^{***}$ (0.037)	$-0.156^{***}$ (0.033)	-0.027 (0.028)	-0.022 (0.022)
Involve			$-0.208^{***}$ (0.037)	$-0.253^{***}$ (0.035)	0.016 (0.025)	0.030* (0.018)
Delay			0.045 (0.075)	0.009 (0.070)	$-0.063^{*} (0.037)$	$-0.055^{*}(0.030)$
Medicaid border			-0.019 (0.045)	-0.012 (0.044)	0.038 (0.047)	0.004 (0.040)
Involve border			$-0.109^{***}$ (0.039)	$-0.077^{**}(0.034)$	$-0.045^{*}\ (0.026)$	$-0.065^{***}$ (0.018)
Delay border			0.015 (0.085)	0.110 (0.076)	$0.063^{*}(0.034)$	0.010 (0.028)
Welfare waiver			0.033 (0.102)	-0.038(0.086)	-0.026 (0.036)	-0.020(0.028)
TANF			0.268** (0.114)	0.222** (0.107)	0.005 (0.160)	0.158 (0.121)
Welfare benefit			0.448*** (0.152)	0.299** (0.132)	0.282 (0.252)	0.115 (0.192)
Family cap			-0.040(0.092)	0.077 (0.077)	-0.044 (0.043)	0.042 (0.033)
DV: ln(abortion rate)	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
CDC/AGI Data	CDC	AGI	CDC	AGI	CDC	AGI
State characteristics	No	No	Yes	Yes	Yes	Yes
State fixed effects	No	No	No	No	Yes	Yes
Year dummies	No	No	No	No	Yes	Yes
State-specific time trends	No	No	No	No	Yes	Yes
$R^2$	0.039	0.047	0.497	0.557	0.912	0.943
State-year combinations	912	912	912	912	912	912

*Notes*: Regression coefficients in this table come from the unbalanced panel of state-year combinations between 1975 and 2005, and are weighted by the female population ages 15–44. Specifically, the estimates are derived from state-years in which CDC and AGI abortion data are both available. These years include 1975–1982, 1984–1985, 1987–1988, 1991–1992, 1995–1996, 2000, and 2005. Standard errors, reported in parentheses, are corrected for arbitrary heteroskedasticity using Huber–White robust standard errors. All models use as the dependent variable the log CDC or AGI abortion rate. All models parameterize the EITC by using the contemporaneous 2 + -child maximum credit. Models (2a) through (3b) control for UR, governor, president, population density, marriage, BA, and females ages 15–19. See Table 1 for an explanation of the variable definitions.

<sup>\*\*\*</sup> Indicate statistical significance at the 0.01 levels.

indicate statistical significance at the 0.05 levels.
 indicate statistical significance at the 0.05 levels.

\* indicate statistical significance at the 0.10 levels.

the incremental benefit for one child and multiple children generate larger abortion responses than the maximum credit. These results are consistent with those reported in Baughman and Dickert-Conlin (2007).

The next several rows investigate changes to the parameterization of the EITC variable, construction of the panel dataset, and the weighting procedure. Generally speaking, the results are robust to these specification checks. Measuring the dependent variable in levels does not substantially change the results. The estimate in column (1) suggests that increasing the EITC maximum credit by \$1000 is associated with a reduction in the abortion rate of about three abortions per 1000 women of childbearing age. This estimate translates to a 15% (3.045/20.94) decrease in the overall abortion rate, a somewhat larger effect than the one from the log model.

This study relies in part on variation in state credits to identify the impact of the EITC. Two concerns are raised by this approach. First, several states structure their EITC programs as nonrefundable tax credits, thereby reducing the likelihood that low-income families will qualify for benefits. Second, given that most states implemented their EITC programs throughout the 1990s, a large portion of this study's observation period contains very little cross-state variation in the EITC.<sup>24</sup> One might therefore question whether the main results are robust to a shortened panel beginning in the early-1990s. To explore these concerns, I create an EITC variable that uses variation in the state credit only if it is structured as a refundable credit. I then test this variable in a regression using state-year combinations between 1990 and 2005. As shown in Table 4, results from this specification check are quite similar to the main results.

Using the balanced panel of state-years does not appreciably change the results. When the six states with missing abortion data (Alaska, California, Louisiana, New Hampshire, Oklahoma, and West Virginia) are omitted from the analysis, the coefficient on the maximum EITC changes only slightly, to a 9% reduction in the abortion rate [column (1)]. Following Bitler and Zavodny (2001), I explore the robustness of the main results to a correction for serial correlation. Indeed, tests reveal that the errors in the abortion regressions are serially correlated within states.<sup>25</sup> To correct for within-state autocorrelation, I use a panel data generalization of the Prais–Winsten method that assumes the autocorrelation follows a common AR(1) process across all states. Column (1) provides autocorrelation-corrected estimates using the log abortion rate as the dependent variable, and column (2) provides autocorrelation-corrected estimates using the log number of abortions. In both cases, the parameter estimate on the EITC is similar to the uncorrected models and continues to be statistically significant. In addition, correcting for the possibility that the errors are contemporaneously correlated across panels (using panel-corrected standard errors) does not appreciably change the precision with which the EITC coefficients are estimated.

As previously stated, one of the drawbacks of the CDC data is that abortion counts are reported by state of occurrence, rather than state of residence. Therefore, to examine the influence of non-state residents crossing state borders to receive an abortion, I estimate models that exclude states with at least 20% and 10% of its abortions performed on out-of-state residents.<sup>26</sup> Results presented in Table 4 suggest that any cross-state abortion migration is not driving the results. Specifically, excluding the 14 states that perform 10% or more of its abortions on non-state residents leads to an EITC effect implying an 8% reduction in the abortion rate and a 9% reduction in the number of abortions. These estimates remain statistically significant despite a 28% reduction in the sample size (from N = 1555 to N = 1123).

As a final specification check, I compare regression estimates derived from CDC and AGI abortion data. Results from this exercise are presented in Table 5. To maintain consistency, all estimates are derived from state-year combinations in which CDC and AGI abortion data are both available.<sup>27</sup> This reduces the sample to 912 state-year combinations. All models use as the dependent variable the log CDC or AGI abortion rate. The AGI data tend to yield coefficients on the EITC and other policy variables that are consistent with those from the CDC data. In models without controls, the coefficient on the EITC implies a 7% and 7.5% reduction in the CDC- and AGI-based abortion rate, respectively. Adding states' observable characteristics leads to similar increases in the magnitude of the EITC effect (CDC estimate: -0.242; AGI estimate: -0.267). Finally, adding the full set of controls for state-level unobserved heterogeneity also leads to similar EITC estimates (CDC estimate: 0.002; AGI estimate: 0.036). Although they are not statistically significant, the positive coefficients in the final model reflect the fact that important gaps appear in the AGI data collection throughout the 1990s and 2000s, a period in which several reforms to the EITC were enacted. For example, AGI abortion counts are reported only three times between 1996 and 2005. During this 10-year period, 10 states implemented their own EITC.<sup>28</sup> Such data gaps therefore reduce substantially the amount of identifying variation in the EITC. With the exception of the coefficients on family caps, CDC- and AGI-based abortion data produce qualitatively similar results for the remaining policy variables. The estimate for family caps is negatively signed when using the CDC data and is positively signed when using the AGI data, although in neither case is the estimate statistically significant.

<sup>&</sup>lt;sup>24</sup> Indeed, 13 of the 17 state EITCs that operated between 1975 and 2005 were implemented on or after 1990. These states are: Colorado (1999, but suspended after 2002), District of Columbia (2000), Iowa (1990), Illinois (2000), Indiana (2002), Kansas (1998), Maine (2000), Massachusetts (1997), Minnesota (1991), New Jersey (2000), New York (1994), Oklahoma (2002), and Oregon (1997). Of the 13 states, 10 structured the program as a refundable tax credit. These states are: Colorado, District of Columbia, Illinois, Indiana, Kansas, Massachusetts, Minnesota, New Jersey, New York, and Oklahoma.

 $<sup>^{25}</sup>$  A Wooldridge (2002) test of no serial correlation in panel data yields a highly significant F-statistic in the log abortion rate (*F* = 57.76) and log number of abortions (*F* = 56.67) regressions.

<sup>&</sup>lt;sup>26</sup> I use CDC (2008) to make these classifications. States excluded because at least 20% of its abortions are performed on out-of-state residents include the following: District of Columbia, Kansas, North Dakota, Rhode Island, and Tennessee. States excluded because at least 10% of its abortions are performed on out-of-state residents include the following: District of Columbia, Kansas, North Dakota, Rhode Island, Arkansas, North Dakota, Rhode Island, Arkansas, Georgia, Nebraska, North Carolina, Oregon, South Dakota, Vermont, and West Virginia.

<sup>&</sup>lt;sup>27</sup> As previously stated, AGI abortion data are available for the following years: 1975–1982, 1984–1985, 1987–1988, 1991–1992, 1995–1996, 2000, and 2005. These data are also reported by the state of occurrence.

<sup>&</sup>lt;sup>28</sup> These states are: Colorado (1999, but suspended after 2002), District of Columbia (2000), Illinois (2000), Indiana (2002), Kansas (1998), Maine (2000), Massachusetts (1997), New Jersey (2000), Oklahoma (2002), and Oregon (1997).

## 6. Conclusion

Using a panel of states between 1975 and 2005, I develop and test a model of the relationship between the EITC and abortion. The theoretical model predicts that the EITC has implications for abortion behavior at two points in the fertility decision-making process: when women are making decisions about the level of sexual activity and contraceptive intensity and after a pregnancy occurs. While the model highlights the possibility that the EITC can increase the number of abortions, the income and price effects associated with the credit suggest that reductions in abortion are highly likely. The empirical results generally support these theoretical propositions, and can be summarized as follows. Increases in the EITC are associated with statistically significant reductions in the overall abortion rate. Main results imply that a \$1000 increase in the EITC maximum credit is expected to reduce the abortion rate by 7.6%. I also find that reductions in abortion are not offset by increases in births. Rather, the EITC appears to be reducing pregnancy rates.

These findings warrant some discussion. Generally speaking, results in this paper are broadly consistent with and may help to explain some counter-theoretical findings in previous studies. In particular, a recent paper by Baughman and Dickert-Conlin (2007) determines that increases in the EITC are associated with small *reductions* in births. Moreover, Kearney (2004) and Levine (2002) find that policies decreasing incremental welfare benefits (through family caps) are associated with *increases* in the birth rate.

One interpretation of the results in this study and Baughman and Dickert-Conlin (2007) focuses on the labor supply effects of the EITC. Previous research finds consistent evidence that the credit is associated with greater employment levels among disadvantaged women (Eissa and Liebman, 1996; Grogger, 2003; Herbst, 2008; Meyer and Rosenbaum, 2001). By increasing the returns to work, however, the EITC raises the opportunity costs associated with bearing and raising children. Such employment effects are important for understanding a key result in this paper: reductions in abortion appear to originate in decisions regarding sexual behavior and contraceptive intensity, as opposed to after a pregnancy occurs. In other words, the channel through which the EITC reduces abortions is a decrease in risky sexual behavior (perhaps through more diligent contraceptive use), and not a shift from abortions to births conditional on being pregnant. Future research should subject this proposition to more rigorous testing, however.

As Baughman and Dickert-Conlin (2007) point out, another interpretation of these results emphasizes Becker's (1991) theory on the demand for children. This model suggests that positive income shocks create trade-offs for families between the number (quantity) and development (quality) of children. If children are assumed to be normal goods, the introduction of an EITC is predicted to increase the demand for children. Alternatively, families might be concerned with the quality of children. In this case, the introduction of an EITC might lead to reductions in higher-order childbearing and an increase in human capital investments for existing children. In this conceptualization, the EITC acts as a mechanism by which parents substitute greater investments in higher-quality children for increases in the quantity of children. Results in this study and others suggest that the latter prediction is more appropriate for explaining fertility responses to the EITC.

Understanding the role of the EITC in shaping women's fertility decisions is important for a number of reasons. Previous studies find that other factors related to fertility—such as abortion legalization and the availability of contraception—have powerful effects on women's lifecycle fertility, marriage decisions, and labor market outcomes (Ananat et al., 2007; Angrist and Evans, 1999; Bailey, 2006; Goldin and Katz, 2002). Policies affecting fertility also have implications for child outcomes, including the propensity to commit crime and use illicit drugs, educational attainment, welfare use, and employment (Ananat, et al., 2006; Charles and Stephens, 2006; Donohue and Levitt, 2001; Gruber et al., 1999). For disadvantaged children, the additional income a family receives from the EITC can be particularly important. Recent studies suggest that the credit lifts 2.7 million children above the poverty line each year, and that EITC-induced increases in family income are associated with gains in children's cognitive test scores (Center on Budget and Policy Priorities, 2005; Dahl and Lochner, 2008). Therefore, by potentially influencing short-run fertility decisions, the EITC has the ability to shape long-term outcomes for adult recipients and their children.

## Acknowledgment

I would like to gratefully acknowledge Laura Peck and two anonymous referees for their helpful comments on previous versions of the paper.

## References

- Akerlof, G., Yellen, J., Katz, M., 1996. An analysis of out-of-wedlock childbearing in the United States. Quarterly Journal of Economics 111, 277–317.
- Ananat, E., Gruber, J., Levine, P., 2007. Abortion legalization and life-cycle fertility. Journal of Human Resources 42, 375–397.

Angrist, J., Evans, W., 1999. Schooling and labor market consequences of the 1970 state abortion reforms. Research in Labor Economics 18, 75–113. Bailey, M., 2006. More power to the pill: the impact of contraceptive freedom on women's life cycle labor supply. Quarterly Journal of Economics 121, 289– 320.

Baughman, R. and Dickert-Conlin, S. (2007). The earned income tax credit and fertility. *Journal of Population Economics*. Forthcoming. Becker, G., 1991. A Treatise on the Family. Harvard University Press, Cambridge, MA.

Decker, G., 1991. A fleatise of the raining. Harvard Officersty Press, Calibridge, MA

Blank, R., George, C., London, R., 1996. State abortion rates: the impact of policies, providers, politics, demographics, and economic environment. Journal of Health Economics 15, 513–553.

Bitler, M., Zavodny, M., 2001. The effect of abortion restrictions on the timing of abortions. Journal of Health Economics 20, 1011–1032.

14

#### C.M. Herbst/Social Science Research xxx (2011) xxx-xxx

Camasso, M., Harvey, C., Jagannathan, R., Killingsworth, M., 1999. New Jersey's family cap and family size decisions: some findings from a 5-year evaluation. Rutgers University, New Brunswick, NJ.

Center on Budget and Policy Priorities, 2005. What Does the Safety Net Accomplish? Center on Budget and Policy Priorities, Washington, DC.

Charles, K., Stephens, M., 2006. Abortion legalization and substance use. Journal of Law and Economics 49, 481–505.

Crouse, G., 1999. State implementation of major changes to welfare policies 1992–1998. <a href="http://aspe.hhs.gov/HSP/Waiver-Policies99/policy\_CEA.htm">http://aspe.hhs.gov/HSP/Waiver-Policies99/policy\_CEA.htm</a>.

Dahl, G., Lochner, L., 2008. The Impact of Family Income on Child Achievement: Evidence from the Earned Income Tax Credit. NBER Working Paper No. 14599. National Bureau of Economic Research, Cambridge, MA.

Dickert-Conlin, S., Chandra, A., 1999. Taxes and the timing of births. Journal of Political Economy 107, 161–177.

Dickert-Conlin, S., Houser, S., 2002. EITC and marriage. National Tax Journal 55, 25-40.

Duchovny, N., 2001. The earned income tax credit and fertility. Unpublished Doctoral Dissertation. College Park, MD: Department of Economics, University of Maryland.

Eissa, N., Liebman, J., 1996. Labor supply to the earned income tax credit. Quarterly Journal of Economics 111, 605–637.

Fang, H., Keane, M., 2004. Assessing the impact of welfare reform on single mothers. Brookings Papers on Economic Activity 35, 1–116.

Finer, L., Henshaw, S., 2006. Estimates of U.S. Abortion Incidence, 2001–2003. Alan Guttmacher Institute, New York, NY.

Goldin, C., Katz, L., 2002. The power of the pill: oral contraceptives and women's career and marriage decisions. Journal of Political Economy 110, 730–770. Grogger, J., 2003. The effects of time limits, the EITC, and other policy changes on welfare use, work, and income among female-headed families. Review of Economics and Statistics 85, 394–408.

Gruber, J., Levine, P., Staiger, D., 1999. Abortion legalization and child living circumstances: who is the marginal child? Quarterly Journal of Economics 114, 263–292.

Herbst, C., 2008. Do social policy reforms have different impacts on employment and welfare use as economic conditions change? Journal of Policy Analysis and Management 27, 867–894.

Horvath-Rose, A., Peters, E., 2001. Welfare waivers and non-marital childbearing. In: Duncan, Greg., Chase-Lansdale, Lindsey (Eds.), Welfare Reform: For Better, For Worse. Russell Sage, New York, pp. 222–244.

Internal Revenue Service (IRS), 2000. Unpublished Analytical Table B for the 1998 tax year.

Joyce, T., Kaestner, R., 1996. State reproductive policies and adolescent pregnancy resolution: The case of parental involvement laws. Journal of Health Economics 15, 579–607.

Joyce, T., Kaestner, R., 2001. The impact of mandatory waiting periods and parental consent laws on the timing of abortion and state of occurrence among adolescents in Mississippi and South Carolina. Journal of Policy Analysis and Management 20, 263–282.

Joyce, T., Kaestner, R., Korenman, S., Henshaw, S., 2004. Family cap provisions and changes in births and abortions. Population Research and Policy Review 23, 475–511.

Kaestner, R., Korenman, S., O'Neill, J., 2003. Has welfare reform changed teenage behaviors? Journal of Policy Analysis and Management 22, 225–248.

Kane, T., Staiger, D., 1996. Teen motherhood and abortion access. Quarterly Journal of Economics 111, 467–506.

Kearney, M., 2004. Is there an effect of incremental welfare benefits on fertility behavior? A look at the family cap. Journal of Human Resources 39, 295–325. Klerman, J., 1998. Welfare reform and abortion. In: Moffitt, Robert. (Ed.), Welfare, The Family, and Reproductive Behavior. National Academy Press, Washington, DC.

Kniesner, T., Ziliak, J., 2002. Tax reform and automatic stabilization. The American Economic Review 92, 590-612.

Levine, P., 2002. The impact of social policy and economic activity throughout the fertility decision tree. Working Paper No. 9021. National Bureau of Economic Research, Cambridge, MA.

Levine, P., 2003. Parental involvement laws and fertility behavior. Journal of Health Economics 22, 861–878.

Levine, P., 2004. Sex and Consequences: Abortion, Public Policy, and the Economics of Fertility. Princeton University Press, Princeton, NJ.

Looney, A., 2005. The Effects of Welfare Reform and Related Policies on Single Mothers' Welfare Use and Employment in the 1990s (Finance and Economics Discussion Series Working Paper 2005–45). Federal Reserve Board, Washington, DC.

Lopoo, L., DeLeire, T., 2006. Did welfare reform influence the fertility of young teens? Journal of Policy Analysis and Management 25, 275–298.

Matthews, S., Ribar, D., Wilhelm, M., 1997. The effects of economic conditions and access to reproductive health services on state abortion rates and birthrates. Family Planning Perspectives 29, 52–60.

Merz, J., Jackson, C., Klerman, J., 1995. A review of abortion policy: legality, medicaid funding, and parental notification, 1967–1994. Women's Rights Law Reporter 17, 1–61.

Meyer, B., Rosenbaum, D., 2001. Welfare, the earned income tax credit, and the labor supply of single mothers. Quarterly Journal of Economics 116, 1063–1114.

Michael, R., 1999. Abortion decisions in the U.S. Harris School Working Paper Series No. 99.6. Harris School of Public Policy Studies, University of Chicago, Chicago, IL.

Moffitt, R., 1998. The effect of welfare on marriage and fertility. In: Moffitt, Robert (Ed.), Welfare, The Family, and Reproductive Behavior. National Academy Press, Washington, DC.

Nagle, A., Johnson, N., 2006. A Hand Up: How State Earned Income Tax Credits Help Working Families Escape Poverty in 2006. Center on Budget and Policy Priorities, Washington, DC.

Offner, P., 2003. Teenagers and Welfare Reform. Working Paper. Urban Institute, Washington, DC.

Phillips, K.R., 2001. Who Know About the Earned Income Tax Credit? Series B, No. B-27. The Urban Institute, Wahington, DC.

Romich, J., Weisner, T., 2000. How families view and use the EITC: advance payment versus lump sum delivery. National Tax Journal 53, 1245–1266.

Rowe, G., Versteeg, J., 2005. Welfare Rules Databook: State TANF Policies as of July 2003. The Urban Institute, Washington, DC.

Scholz, J., 1994. The earned income tax credit: participation, compliance, and antipoverty effectiveness. National Tax Journal 47, 63–87.

Sollom, T., Various Years. State actions on reproductive health issues. Family Planning Perspectives, various editions.

Turturro, C., Benda, B., Turney, H., 1997. Arkansas Welfare Waiver Demonstration Project, Final Report (July 1994 through June 1997). University of Arkansas at Little Rock, School of Social Work, Little Rock, AR.

Urban-Brookings Tax Policy Center, 2008. Spending on the EITC, Child Tax Credit, and AFDC/TANF, 1976–2010. Urban-Brookings Tax Policy Center, Washington, DC.

US Department of Health and Human Services (DHHS), 1997. Setting the Baseline: A Report on State Welfare Waivers. <a href="http://aspe.hhs.gov/HSP/isp/waiver2/title.htm">http://aspe.hhs.gov/HSP/isp/waiver2/title.htm</a>>.

US General Accounting Office, 1997. Welfare Reform: States' Early Experiences with Benefit Termination. Report No. HEHS-97-74. US General Accounting Office, Washington, DC.

US House of Representatives, Committee on Ways and Means, 2004. Green book, background material and data on programs within the jurisdiction of the Committee on Ways and Means. Government Printing Office, Washington, DC.

Wang, X., Chen, C., Wang, L., Chen, D., Guang, W., French, J., 2003. Conception, early pregnancy loss, and time to clinical pregnancy: a population-based prospective study. Fertility and Sterility 79, 577–584.

Whittington, L., 1992. Taxes and the family: the impact of the tax exemption for dependents on marital fertility. Demography 29, 215-226.

Whittington, L., Alm, J., Peters, E., 1990. Fertility and the personal exemption: implicit pronatalist policy in the United States. American Economic Review 80, 545–556.

Wilcox, A., Baird, D., Weinberg, C., 1999. Time of implantation of the conceptus and loss of pregnancy. New England Journal of Medicine 340, 1796–1799. Wooldridge, J., 2002. Econometric Analysis of Cross-Section and Panel Data. MIT Press, Cambridge, MA.